Answer to Reviewer #1:

1. A review about Chikungunya fever can contribute to a better understanding of the disease, its symptoms, transmission, and treatment options. By summarizing and synthesizing existing research on CF, this review will provide an overview of the current state of knowledge and identify gaps in the research that need to be addressed.

The authors would like to thank the reviewer for his considerations and recognition of their work.

2. Please provide further insights into the effectiveness of different treatment approaches and prevention measures, as well as highlight the most promising areas for future research. By providing a comprehensive and up-to-date overview of the literature, this review will help inform policymakers, healthcare professionals, and the general public about the most effective ways to prevent and treat CF.

The request was fulfilled in the text of the manuscript.

3. Persistent arthralgia is a significant complication of Chikungunya fever, as it can affect a person’s quality of life and ability to carry out daily activities. The exact mechanisms behind persistent arthralgia are not fully understood, but it is thought to be caused by a combination of factors, including inflammation, autoimmunity, and tissue damage. Please make a detailed introduction to this part.

The request was fulfilled in the text of the manuscript.

4. Studies have shown that persistent arthralgia can affect up to 60% of individuals who have had Chikungunya fever, and it can last for several years in some cases. Treatment options for persistent arthralgia associated with Chikungunya fever include pain management, physical therapy, and in some cases, disease-modifying antirheumatic drugs (DMARDs). Please provide medical advice to manage the symptoms effectively.

The request was fulfilled in the text of the manuscript.

5. Please add Conclusions to this review.

The request was fulfilled in the text of the manuscript.